Author's response to reviews

Title: Self-care coping strategies in people with diabetes: a qualitative exploratory study

Authors:

Margaret M Collins (MCollins@co.tuolumne.ca.us)
Colin P Bradley (c.bradley@ucc.ie)
Tony O'Sullivan (tonyosu@eircom.net)
Ivan J Perry (i.perry@ucc.ie)

Version: 3 Date: 20 January 2009

Author's response to reviews: see over
January 20, 2009

Natalie Pafitis MSc
Senior Assistant Editor
BMC-series Journals
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com

Dear Natalie,

We are delighted with the acceptance of 1555539498193412 - Self-care coping strategies in people with diabetes: a qualitative exploratory study paper. We were very pleased with the review process, and felt the reviewer comments were helpful and enhanced the final manuscript. Attached please find a copy of our revised manuscript. A section on ethics approval has been added to the methods section as requested.

Kind Regards,

Professor Ivan J. Perry, MD, PhD